Author's response to reviews

Title: Associations between diet and disease activity in ulcerative colitis patients using a novel method of data analysis

Authors:

Elizabeth A Magee (EMagee@dundee.ac.uk)
Laurie M Edmond (L.M.Edmond@dundee.ac.uk)
Shiona M Tasker (s.m.tasker@dundee.ac.uk)
San Choon Kong (s.c.kong@dundee.ac.uk)
Richard Curno (r.curno@dundee.ac.uk)
John H Cummings (J.H.Cummings@dundee.ac.uk)

Version: 6 Date: 9 February 2005

Author's response to reviews:

Just changes to previous web page (manuscript details). Cover letter provided with previous upload. Sorry I'm making a mess of this.